A THESIS FOR THE DEGREE OF MASTER OF SCIENCE

Ultrasonic Measurements of Blood and Plasma coagulation



Sitaramanjaneya Reddy Guntur

Department of Marine Instrumentation Engineering

GRADUATE SCHOOL CHEJU NATIONAL UNIVERSITY

Ultrasonic Measurements of Blood and Plasma coagulation

Sitaramanjaneya Reddy Guntur

(Supervised by Professor Dong Guk Paeng)

A thesis submitted in partial fulfillment of the requirement for the degree of Master of Science

> 2005.12. This thesis has been examined and approved

Thesis director, Jin ho Bae, prof. of Marine Instrumentation Engineering

Min Joo Choi, Prof. of Biomedical Engineering

Dong Guk Paeng, Prof. of Marine Instrumentation Engineering

2005.12. Date

Department of Marine Instrumentation Engineering

GRADUATE SCHOOL CHEJU NATIONAL UNIVERSITY

TABLE OF CONTENTS

| •••••• | iii |
|---|--------|
| ABSTRACT | v |
| LIST OF TABLES | vii |
| LIST OF FIGURES | viii |
| CHAPTER I INTRODUCTION | 1 |
| 1.1 General | 1 |
| 1.2 Previous Studies | |
| 1.2.1 Whole blood during coagulation | |
| 1.2.2 Plasma during coagulation with aPTT | |
| 1.2.3 Plasma during coagulation with aPTT and her | oarin7 |
| 1.3 Specific Aims of this Research | 8 |
| 1.4 Overview | 9 |
| CHAPTER II THEORY FOR MEASUREMENT OF | SOUND |
| SPEED AND ATTENUATION COEFFICIEN | |
| 2.1 Introduction | |
| 2.2 Ultrasonic velocity | |
| 2.3 Ultrasonic attenuation | |
| 2.4 Reflection and Transmission of Plane Waves Normally | |
| Incident on a Layer between two Media | 14 |
| CHAPTER III MATERIALS AND METHOD | |
| 3.1 Introduction | 18 |
| 3.2 Properties of Blood | |
| 3.2.1 General | |
| 3.2.2 Blood and plasma samples | |
| 3.2.3 Chemicals and other materials | |
| 3.3 Measurement procedure | |
| CHAPTER IV MEASUREMENT OF ACOUSTIC PROPI | ERTIES |
| OF WHOLE BLOOD DURING COAGULAT | |
| 4.1 Introduction | |
| 4.2 Blood preparation | |
| 4.3 Method for whole blood coagulation measurement | |
| 4.4 Results | |

i

| 4.5 Disc | cussion | 37 |
|------------|--|------------|
| | clusion | |
| CHAPTER V | MEASUREMENT OF ACOUSTIC PROPER' OF PLASMA DURING COAGULATION | |
| 5 1 Intro | oduction | |
| | ma preparation | |
| | hod for plasma coagulation measurement | |
| | ults | |
| | cussion | |
| 5.6 Con | clusion | 52 |
| CHAPTER VI | MEASUREMENT OF ACOUSTIC PROPER' OF PLASMA DURING COAGULATION W HEPARIN | ITH |
| | | |
| | oduction | |
| | hod for plasma coagulation with heparin measurement ults | |
| | cussion | |
| | clusion | |
| CHAPTER VI | I CONCLUSIONS AND FUTURE WORK | 63 |
| 7.1 Con | clusion | 63 |
| | ire work | |
| BIBLIOGRAI | PHY 제주대학교 중앙도서관 | 65 |
| APPENDIX | OF PLASMA DURING COAGULATION W | ITH |
| | BROWN AND RED ALGE | |
| | oduction | |
| | terials and Method | |
| | ults and Discussion | |
| | DGEMENTS | |
| ACANUWLL | | /ð |

ii



가

.

,

.

iii



.

가

. ,

iv

ABSTRACT

The haemostatic system has a dual paradoxical function in the body. It should stop bleeding whenever needed, but also keep the blood flowing in the circulatory system without any obstructing blood clots. The system is complex with series of enzymatic reactions, known as the coagulation cascade and intertwined processes that interact to produce a fine-tuned regulation of the performance. Abnormal concentration of one or more factors of blood coagulation results in an inadequate formation of fibrin by excessive coagulation ability, thrombophilia, and haemostatic disorders. Therefore, the main focus is to measure the whole process of coagulation using ultrasound.

The aims of this work are to develop an ultrasonic method for analyzing the whole process of blood and plasma coagulation and anti-coagulant activity of heparin, and to investigate the physiological phases of haemostatic system in terms of acoustic properties. To achieve this goal an ultrasonic pulse-echo method has been used for measurement of sound speed and attenuation coefficient in blood and plasma samples with a center frequency of 5 MHz at temperature of 37°C. The experiments were carried out using ten samples of human blood as an average hematocrit of 41 %. Plasma coagulation with aPTT (Activated Partial Thromboplastin Time) and with an anticoagulant has also been evaluated for haemostatic studies using Lab view program in real time.

Experimental results demonstrate that the average of sound speed and attenuation coefficient of whole blood were increased from 1584 to 1595 m/s, 2.2 to 4.3 dB/cm respectively, while the ones in plasma were increased from 1535 to 1541 m/s, 1.25 to 3.2 dB/cm during coagulation. The slope of

V

sound speed and attenuation of blood and plasma was changed at different physiological phases of enzymatic, propagation, and termination phases. Plasma coagulation with an anti-coagulant such as heparin was found to change the acoustic properties less with different patterns. In conclusion the combination of ultrasonic measurements of blood and plasma coagulation with/without an anticoagulant might help for a better understanding of coagulation processes, which may be applied in various techniques for the evaluation of the effects and development of anticoagulants and the drugs.



vi

LIST OF96666999 TABLES

| Table 1 Average sound speed and attenuation from eight samples during | |
|---|----|
| blood and plasma coagulation and comparison with the previous | |
| groups | 30 |



vii

LIST OF FIGURES

| 1.1 | Coagulation cascade pathway (Guyton 2000) | 2 |
|-----|--|----|
| 1.2 | Physiological phases of blood coagulation process | 3 |
| 2.1 | Time shift mechanism between wrap layers and reflected signal | |
| | with and without sample | 11 |
| 2.2 | Amplitude difference of reflected signal from perfect reflector with | |
| | and without sample | 16 |
| 2.3 | Through transmission technique | 16 |
| 3.1 | Sample container with an acoustic window | 21 |
| 3.2 | Ultrasonic measurement system | 23 |
| 4.1 | Block diagram of whole blood coagulation process | 27 |
| 4.2 | Sound speed variations of eight individuals during blood | |
| | coagulation | 31 |
| 4.3 | Average sound speed variation of eight individual during blood | |
| | coagulation | 32 |
| 4.4 | coagulation | |
| | coagulation | 33 |
| 4.5 | Attenuation coefficient variation of eight individuals during blood | |
| | coagulation | 34 |
| 4.6 | Average attenuation coefficient variation of eight individuals | |
| | during blood coagulation | 35 |
| 4.7 | Attenuation variation of nine and ten individual during blood | |
| | coagulation | 36 |
| 5.1 | Block diagram of plasma coagulation process | 41 |
| 5.2 | Sound speed variation of eight individuals during plasma | |
| | coagulation | 44 |

viii

| 5.3 | Average sound speed variation of eight individuals during plasma | |
|-----|--|----|
| | coagulation | 45 |
| 5.4 | Sound speed variation of nine and ten individual during plasma | |
| | coagulation | 46 |
| 5.5 | Attenuation coefficient variation of eight individuals during plasma | |
| | coagulation | 47 |
| 5.6 | Average attenuation coefficient variation of eight individuals | |
| | during plasma coagulation | 48 |
| 5.7 | Attenuation variation of nine and ten individual during plasma | |
| | coagulation | 49 |
| 6.1 | Block diagram of plasma coagulation with heparin process | 55 |
| 6.2 | Sound speed variation of five individuals during plasma | |
| | coagulation with heparin | 57 |
| 6.3 | Average sound speed variation of eight individuals during plasma | |
| | coagulation with heparin | 58 |
| 6.4 | Attenuation coefficient variation of five individuals during plasma | |
| | coagulation with heparin | 59 |
| 6.5 | Average attenuation coefficient variation of five individuals during | |
| | plasma coagulation with heparin | 60 |
| 8.1 | Sound speed variation of five individuals in plasma with brown | |
| | algae during coagulation | 73 |
| 8.2 | Average sound speed variation of five individuals during plasma | |
| | coagulation with brown algae | 73 |
| 8.3 | Attenuation variation of five individuals during plasma coagulation | |
| | with brown algae | 74 |
| 8.4 | Average attenuation variation during plasma coagulation with | |
| | brown algae | 74 |

ix

| 8.5 | Sound | speed | variation | of | five | individual | s during | plasma | |
|-----|---|-----------|--------------|-------|---------|--------------|------------|----------|-----|
| | coagula | tion with | h red algae | ••••• | | | | | .75 |
| 8.6 | Average | e sound | speed var | iatio | n of fi | ive individu | als during | g plasma | |
| | coagula | tion wit | h red algae | | | | | | .75 |
| 8.7 | Attenuation variation of five individuals during plasma coagulation | | | | | | | | |
| | with rec | l algae | | | | | | | .76 |
| 8.8 | Average | e attenua | ation of fiv | e in | dividu | als during p | olasma coa | gulation | |
| | with rec | l algae | | | | | | | .76 |



Х

Chapter I

Introduction

1.1 General

Thrombotic conditions affect more then 600,000 people in the world (Brown 1984). This means their blood has a tendency to clot without any injury (Brown 1984). Due to their conditions, patents will have a need for monitoring either during cardiac surgical procedure or on a regular basis during anticoagulant therapy.

Blood coagulation is a complex physiological process that helps to maintain hemostasis within the vascular system. It should stop bleeding whenever needed, but also keep the blood flowing in the circulatory system without any obstructing blood clots. It consists of series of enzymic reactions following activation of blood clot mechanism. There is a delicate balance among blood cells, coagulation factors and external tissue factors. The activation of these series of enzymatic reactions is known as coagulation cascade (Guyton 2000) as shown in Figure 1.1. These reactions are typically catalyzed by 20 different substances, most of which are plasma glycoproteins, with each step initiating another reaction until fibrin formation (Brown 1984). Abnormal concentration of one or more factors of blood coagulation results in an inadequate formation of fibrin.

According to physiological processes, blood coagulation is characterized by three phases, enzymic phase, propagation phase, and termination phase (Mann 1992) as shown in Fig 1.2. Extrinsic factor Xase is a principle player



Figure 1.1 Coagulation cascade pathway (Guyton 2000)



Enzymatic phase



Propagation phase



Termination phase

Figure 1.2 Physiological phases of blood coagulation process (Mann 1992)

of enzymic phase, in which a tiny amount of thrombin is produced and prothrombin activator catalyzes the conversation of prothrombin to thrombin. Vigorous thrombin generation occurs during propagation phase in intrinsic Xase and continues reaction with prothrombinase (Mann 2002), starting to cleave fibrinogen and activate the factor XIII. Termination phase occurs through the stoichiometric inhibitors antithrombin III (AT-III) and tissue factor pathway inhibitor (TFPI) and through the protein C system. The clotting time is considered to mark the end point of enzymatic phase and the onset of propagation and termination phases.

Coagulation process is the primary stage that comprises the development of anti-coagulant drugs in medical industries. More over, the coagulation process is often used as a reference for determining the activity of anticoagulants. Therefore, an accurate determination of coagulation process is very important for an optimal evolution of anticoagulation drugs. On the other hand, abnormalities in the platelets, coagulation factors and blood vessel result in excessive bleeding. Therefore measurement of blood coagulation process may benefit to understand both types of conditions to improve diagnostic and therapeutic process.

1.2 Previous Studies

In last decades, monitoring of blood and plasma coagulation has been carried out in order to determine the coagulation process or/and clotting time. The investigations are based on electrical (Miale 1965), optical (Davey and Malley 1972), and laser (Pieiere et al 2004) methods. These systems were unable to measure the whole process of coagulation. Therefore, an ultrasonic method was introduced for monitoring the change of whole process of blood coagulation (Shung et al. 1984; Volesis 2002).

Measurement of acoustic properties of whole blood and plasma may allow us to detect the characteristics of the blood clotting deficiencies related to various diseases, to analyze anticoagulant activity of anticoagulant, to improve diagnosis of some diseases and to suggest the appropriate treatment.

1.2.1 Whole blood during coagulation

The first digital ultrasonic interferometer was introduced for testing small amount of whole blood sample during coagulation (Grybauskas et al. 1978). They investigated velocity and absorption of whole blood during coagulation. Further, ultrasonic quantitative parameters, including sound speed and attenuation were measured, and they have been demonstrated to be useful in the sensitive detection of the process of coagulation (Shung et al. 1984; Voleisis et al. 2002; Wang et al. 2002). In addition, high frequency ultrasound to measure the quantitative back scattering parameters and acoustic velocity as a function of hematocrit percentage of whole blood during coagulation were measured (Ossant et al. 2004). These previous researchers measured the whole blood coagulation at various temperatures using citrated blood with an addition of calcium chloride for coagulation. They did not investigate whole blood coagulation in real body temperature, as natural processes of coagulation. Previous measurements of sound speed and attenuation on whole blood coagulation have shown that coagulation process may be affected at different temperatures conditions, with different hematocrit percentage, using citrated whole blood coagulation with thrombin or calcium chloride (shung et al.1984; Ossant et al. 2004; Huang et al. 2005). To consider all the previous researches, it is crucial to develop another feasible approach to measure the whole blood coagulation at

different hematocrit percentages with a natural process of coagulation at real body temperature.

1.2.2 Plasma during coagulation with aPTT

The determination of plasma coagulation time is an essential part of monitoring therapeutic anticoagulants. Standard methodologies for the measurement of plasma clotting are available to evaluate elapsed time from the beginning of an induced coagulation process up to the formation of a stable fibrin polymer. Currently, accepted methodologies are based on optical and mechanical. Optical detection can be preformed either with naked eyes or as a reduction in transmittance when light passes through the sample (Davey and Malley 1972). Mechanical transduction is typically carried out with a vibrating or rotating metal ball. Cessation of ball moving was considered as the endpoint of coagulation. An ultrasonic method has been studied from last few years to measure the plasma clotting time. Alves and Machado used acoustic streaming method using backscatter signals from spherical glass particles (in 1991) and using ultrasonic shear wave method (in 1994) to detect the plasma clotting time.

Deviation from normal time range of plasma clotting of previous researches indicating that some factors is not adequately performing its function in the process, and further tests will be required for its identification. Each test initiates the coagulation process at a different stage and the measured time provides clue about the malfunction factor. An ultrasonic method to measure whole coagulation process has been employed for the monitoring of plasma coagulation. During plasma coagulation, the coagulum changes due to the formation of a fibrin clot. In turn, this changed coagulum shifts the characteristic resonance amplitude of the ultrasonic

transducer enabling real-time continuous monitoring of this biological event. By monitoring the signal output as a function of time, a distinct blood clotting profile can be seen.

1.2.3 Plasma during coagulation with aPTT and heparin

The activated partial thromboplastin time (aPTT) and the activated clotting time (ACT) are the most extensively applied measurements of the heparin effect. Coagulation analyzer (such as the Hemochron system) is the principle player of measurement of aPTT and ACT. Various new and improved coagulation analyzers (devises) were available on market in recent years (Cheng et al. 1998; Hepel et al. 2003). Hand held prototype of small capillary coagulation analyzer have been recently developed. They are based on the drawing of blood or plasma into small capillary, and coagulation is detected using either a laser (biotrack, ciba-corning) or pressure sensing system (Nyco Med). However these systems are quite complex and require a laser or pressure sensing equipment. The other method was a piezoelectric quartz crystal (PQC) sensor for determination of blood and plasma coagulation with heparin (Chang et al. 2000; Cheng et al. 1998). None of the previous studies investigated the properties of heparin during coagulation with plasma using ultrasound. In order to investigate the whole process of plasma coagulation with heparin as a function of time, ultrasonic pulse echo method was used in this study. This study investigated the whole process of plasma coagulation with heparin and effects of specific factors and/or process during coagulation in terms of sound speed and attenuation.

1.3 Specific Aims of this Research

This research primarily is to measure ultrasonic velocity and attenuation coefficient measurement of whole blood and plasma during coagulation. Specifically, the studies can be divided into three aims. The first aim is to measure the acoustic properties of whole blood during coagulation at real body temperature with natural process when exposed to the air. The acoustic properties of whole blood during coagulation at body temperature of 37°C are needed due to the increasing interest to identify the three different phases of coagulation as indicated in physiology. Human blood coagulation at real body temperature, at a certain hematocrit and natural process without any chemicals may be higher sensitivity than at the room temperature with addition of calcium chloride.

The second aim is to thoroughly investigate the whole process of plasma coagulation with aPTT. Until now no investigations are carried out to measure the whole process of plasma coagulation. A feasible method for measurement of acoustic properties of plasma during coagulation with respect to time was developed. Structural variation of whole process of coagulation was investigated since the standard clinical method may not be applicable to know each stage of coagulation.

The third aim is to extend the study to the plasma coagulation with commercial available heparin, to study the effects of specific factors and/or process of heparin as an anticoagulant using ultrasound. The ultrasonic properties of plasma during coagulation with heparin may identify coagulation/clotting time from the ultrasound signals.

1.4 Overview

There are six chapters in addition to the chapter of introduction.

Chapter 2 provides the theoretical background for measurement of ultrasonic velocity and attenuation coefficient from both blood and plasma during coagulation. Attenuation coefficient measurement model includes three layer transmission techniques to interpret the attenuation in time domain.

Material and Methods for measurement of acoustic properties of whole blood and plasma during coagulation was presented in Chapter 3. Properties of both blood and plasma were introduced briefly. Measurement of acoustic properties of blood and plasma coagulation using a pulse echo method was discussed in detail.

Chapter 4 describes the measurement of acoustic properties of whole blood during coagulation and analysis of three distinguishable phase of blood coagulation in terms of sound speed.

Chapter 5 shows the results on measurement of sound speed and attenuation coefficient of plasma during coagulation with aPTT. Analysis of whole process of coagulation was discussed.

Chapter 6 covers the measurement of sound speed and attenuation coefficient of plasma coagulation with heparin. Specific factors and process of plasma coagulation with heparin in terms of sound speed and attenuation were investigated.

Chapter 7 summarized the findings in this research and also provides some recommendations for future work.

At last, appendix includes the preliminary experiments with other anticoagulants (brown and red algae) from seaweed were presented.

Chapter II

Theory for measurement of sound speed and attenuation coefficient

2.1 Introduction

The theories about sound speed and attenuation coefficient were described in this chapter for a better understanding the experiments on blood and plasma coagulation with/without heparin. It includes the theory about the change of sound speed and attenuation coefficient within the blood/plasma sample during coagulation. Sound speed and attenuation coefficient are important ultrasonic parameters to derive blood and plasma coagulation constants and characterization of whole process of coagulation.

2.2 Ultrasonic velocity

The technique used to measure ultrasonic velocity of blood /plasma during coagulation was classical differential pulse-echo method. A change in sound pulse transit time when a sample is incepted with in the path was measured by using known speed of sound to infer the sound speed of the sample. Consider a source of an ultrasonic pulse S, and a received ultrasonic pulse R as shown in Fig.2.1. Suppose a right rectangular sample container of homogeneous blood/plasma sample is inserted in the path of beam, such that its flat faces are perpendicular to the direction of propagation of the pulse. The dotted box in Fig. 2.1 is defined as thickness of the container (L_m). Triggering is done via the pulse generated at the source so that the received pulse displayed on an oscilloscope looks like that as shown in Fig.2.1.



Figure 2.1 Time shift mechanism between wrap layers and reflected signal with and without sample

The time shift (t_2-t_1) in the position of the RF pulse from sample the front and rare face and the time shift (Tw-Tm) from the steel reflector with and without the sample in the path (Kuo 1990) as shown in Fig .2.1.

$$L_{m} = L - L_{1} - L_{2} = L - (t_{1}/2) C_{w} - [(T_{m} - t_{2})/2] C_{w}$$
(2.1)

$$T_w/2 = L/C_w = 1/C_w (L_1 + L_m + L_2)$$
(2.2)

$$T_m/2=1/C_w(L_1+L_2)+L_m/C_m$$
 (2.3)

From Equation (2.2) and (2.3)

$$T_m - T_w/2 = L_m (1/C_m - 1/C_w)$$
 (2.4)

Substituting Equation (2.1) into (2.4), we have

$$T_{m} T_{w} / 2 = (T_{w} / 2 * C_{w} - t_{1} / 2 * C_{w} - (T_{m} - t_{2}) / 2 * C_{w}) * (1 / C_{m} - 1 / C_{w})$$
(2.5)
efore

Therefor

$$C = Cw [((Tw-Tm)/(t_2-t_1)) + 1]$$
(2.6)

, where C_w is the sound speed of water, T_w and T_m are the time of flight from the transducer to the reflector without the sample and with sample, respectively. t_1 and t_2 are the time from the front and the rear face of the sample, respectively, and C is the sound speed of blood or plasma sample.

2.3 Ultrasonic attenuation

The amplitude of a plane ultrasonic wave decreases as the propagating distance increases. The amplitude of the ultrasonic wave which has propagated the distance of x is represented as

$$A(X) = A_0 e^{-\alpha x} \tag{2.7}$$

, Where α is the attenuation coefficient of the propagation media. X is distance traversed. A_0 is the attenuation of the wave and A is the amplitude after a distance x has been traversed. Like the sound speed, the attenuation coefficient is dependent upon the physical properties of the medium. However, unlike the sound speed, the attenuation coefficient varies greatly with frequency. It can be write as $\alpha = \alpha_a + \alpha_s$. Where α_a is the absorption coefficient and corresponds to the conversion of wave energy into internal energy (heat) of the medium. α_s corresponds to the scattering of the primary beam energy into all the different directions. The dimension of α is length ⁻¹, and it can be expressed in nepers/cm. In medical acoustics it is more common to express the attenuation coefficient in dB/cm. To measure the attenuation coefficient as a function of total transmition coefficient of blood and plasma, the amplitude of the reflected echoes with and without sample were compared (Madsen et al.1999), as shown in Fig. 2.2.

$$\alpha = (20 / d) \log_{10} [A_0 / (A / T_{total})]$$
(2.8)

, where A_o and A are the peak to peak signal amplitude without and with the sample, respectively.

2.4 Reflection and Transmission of Plane Waves Normally Incident on a Layer between two Media

When a plane wave is normally incident on the plane interface, the wave incident on the front and rear surface of a sample container is reflected, and the reflected wave returned into the transducer, and some of it propagate and reflect from a perfect reflector. The wave from the reflector meets the boundaries and reflects and transmits repeatedly.

$$R = \frac{r_2 - r_1}{r_2 + r_1} \tag{2.10}$$

1+R=T, we have

$$T = \frac{2r_2}{r_2 + r_1} = \frac{2r_2 / r_1}{r_2 / r_1 + 1}$$
(2.11)

The intensity transmission and reflection coefficients, T_I and R_I , respectively, are in the followings (Kinsler et al. 1980, pp. 124-131, p. 462),

$$R_{I} = I_{r} / I_{i} = |R|^{2}$$
(2.12)

$$T_{I} = I_{t} / I_{i} = (r_{1} / r_{2}) |T|^{2}$$
(2.13)

$$R_{I} = \left(\frac{r_{2} - r_{1}}{r_{2} + r_{1}}\right)^{2} = \left(\frac{r_{2} / r_{1} - 1}{r_{2} / r_{1} + 1}\right)^{2}$$
(2.14)

$$T_{I} = \frac{4r_{2}r_{1}}{(r_{2}+r_{1})^{2}} = \frac{4r_{2}/r_{1}}{(r_{2}/r_{1}+1)^{2}}$$
(2.15)

Where r_1 is the characteristic acoustic impedance of the medium in which the incident and reflected wave are traveling, and r_2 is that of the medium in which the transmitted wave is traveling. In ultrasonic testing, Eqs. (2.12) and (2.13) are generally used to obtain intensity reflection and transmission coefficients between a transducer and a test material. Since the case of the material being tested in this laboratory, a thin (100 µm) layer of OHP (Overheard Projector) film was used in this experiment for an acoustic window and this thin layer should be considered to compute the intensity reflection and transmission coefficients as shown in Figure 2.3.





Figure 2.2 Amplitude difference of reflected signal from perfect reflector with and without sample



Figure 2.3 Through transmission technique

When a layer of finite thickness is formed between two media, an ultrasonic wave normally incident on the interface from medium 1 generates the reflected waves and transmitted waves into the layer. The transmitted wave into the layer is reflected again at the rear interface and the transmitted into medium 3. In this case, the reflection coefficient R is expressed as follows

$$R = \frac{(1 - r_1 / r_2) \cos k_2 L + j(r_2 / r_3 - r_1 / r_2) \sin k_2 L}{(1 + r_1 / r_3) \cos k_2 L + j(r_2 / r_3 + r_1 / r_2) \sin k_2 L}$$
(2.16)

The intensity transmission coefficient, T_I is represented as follows:

$$T_{I} = 1 - |R|^{2}$$
(2.17)

$$T_{I} = \frac{4}{2 + (r_{3}/r_{1} + r_{1}/r_{3})\cos^{2}k_{2}L + (r_{2}^{2}/r_{1}r_{3} + r_{1}r_{3}/r_{2}^{2})\sin^{2}k_{2}L}$$
(2.18)

Here

$$k_2 = \frac{2\pi f}{C_2} = \frac{2\pi f \rho_2}{r_2}$$
(2.19)

$$T_{total} = \frac{4}{2 + (r_3/r_1 + r_1/r_3)\cos^2(2\pi f \rho_2 L/r_2) + [r_2^2/(r_1 r_3) + (r_1 r_3)/r_2^2]\sin^2(2\pi f \rho_2 L/r_2)}$$
(2.20)

, where T_{total} is the total amplitude transmission coefficient through both bounded wrap layers. r_1 , r_2 , and r_3 are the acoustic impedance of water, the OHP film and the blood, respectively. *f* is frequency, ρ_2 is the OHP film density and L is the OHP film thickness.

Chapter III

Materials and Method

3.1 Introduction

The physical properties of blood and plasma will be briefly described in this chapter. There are various approaches for the measurement of sound speed and attenuation coefficients of whole blood and plasma during coagulation. The standard pulse echo method has been used for a long time to measure the quantitative backscattering, includes sound speed and attenuation coefficient of whole blood during coagulation as a function of time or frequency (Shung et al. 1984; Voleisis et al. 2002; Wang et al. 2002). Measurement of whole blood coagulation at real body temperature, at a certain hematocrit and natural process without any chemicals, whole process of plasma coagulation with /without heparin in terms of sound speed and attenuation coefficient has been proposed. The pulse echo method was discussed in this chapter.

3.2 Properties of Blood

3.2.1 General

Blood is the fluid of life, transports oxygen from the lungs to body tissue and carbon dioxide from body tissue to the lungs. Blood is the fluid of growth, transporting nourishment from digestion and hormones from glands throughout the body. Blood is the fluid of health, transports disease fighting substances to the tissue and waste to the kidneys. It is mainly composed of about ~ 45 % of red blood cells (erythrocytes) and ~ 55 % of viscous fluid called plasma with small amount of white blood cells (leukocytes) and platelets (thrombocytes). Plasma is basically a saline solution of three major proteins including fibrinogen, globulin and albumin. The volume of red blood cells in the whole blood is defined as hematocrit and it is expressed in percentage.

Plasma contains clotting factors such as factor VIII, which is used in the treatment of hemophilia, albumin used for the treatment of surgical shock or severe burns, and Immunoglobulins, for protection of the body against infectious diseases and the treatment of immunodeficency. However, it must be noted that blood plasma consists of more than 90 % water. Albumin (the major component), Immunoglobulins, fibrinogens and other components used in the coagulation process. Other components are also important but the fractions are small, like lipids less then 1 gram per 100 milliliters.

3.2.2 Blood and plasma samples

Experiments were conducted using 7 ml of whole blood and 2 ml of plasma sample of ten individual healthy volunteers, without any history of bleeding or thrombosis. The blood was drawn from healthy volunteers in 20 ml syringe and a 7ml of the blood was immediately transferred to the container for measuring the acoustic properties of whole blood. The other 9 ml was placed into 1 ml of 3.8% sodium citrate solution for preparation of plasma. Before preparing plasma sample hematocrit was measured using a centrifuge (HA-200, Hanil Science Industrial). In the laboratory, plasma from blood was prepared by centrifugation of blood for 15 min at about 3500 RCF (relative centrifugal force) using a (Hanshin Medical Co. Ltd., Model-AT-650) centrifuge and than plasma was stored in -70°C until assayed.

3.2.3 Chemicals and other materials

Activated partial thromboplastin reagent aPTT (ellagic+ bovine phospholipids) and calcium chloride (CaCl₂) of 0.02M were obtained from an international reagent corporation (Japan).Heparin was obtained from a commercial pharmacy. The blood sample container was designed using acryl material of dimensions 10 cm long x 2 cm wide x 2 cm thick and in that 2 cm wide and 4 cm drill was made in the front and back side of sample container and sealed with OHP film of 100 micron thick for acoustic window as shown in Fig 3.1





3.3 Measurement procedure

Measurements were made with the experimental configuration depicted in Fig 3.2. The experiments were performed in a degassed water tank at temperature of 37.0±0.2°C. A temperature controller with a liquid circulator was used for keeping temperature constant, and temperature was continuously monitored by an immersed thermometer.

First, the acoustic impedance and sound speed of OHP film were computed using a three-layer transmission technique (Kingsler and Frey), film density and thickness were obtained from manufacturer. The attenuation coefficient of water was assumed to be negligible (Madsen et al. 1999). The sound speed and attenuation coefficient of the blood and plasma samples were determined by a pulse echo method using a Pulser/receiver (Panametrics 5800 PR, CA, USA). Blood sample was placed in the focal zone of a 5 MHz transducer (*Panametrics V326*) with a diameter of 9.5 mm and the sample was positioned perpendicular to the direction of propagation of ultrasound. Ultrasonic pulses were periodically transmitted at a pulse repetition frequency of 500 Hz. The distance between the transducer and the front surface of the sample was approximately 13 cm was maintained to minimize diffraction error. A magnetic stirrer at the bottom of the tank prevented the blood settling. The reflected acoustic signals from the front and back surface of the sample container and a steel reflector were digitized at a sampling rate of 250 Megasamples/s and recorded in a LeCroy (LC547AL USA) digital oscilloscope. The sound speed and attenuation of blood and plasma samples were determined by comparing the reflected signals from the reflector with and without the sample in place.



Figure 3.2 Ultrasonic measurement system

After coagulation, mass and volume of the coagulated blood or plasma sample were measured to find the density and acoustic impedance of blood or plasma. The experimental data were collected every 4 seconds over 65 minutes in blood coagulation and over 45 minutes in plasma coagulation. Collected data were analyzed using MATLAB[®]. Moving average was taken on every 15 data points of the stored data to smoothen the profiles of blood and plasma.


Chapter IV

Measurement of acoustic properties of whole blood during coagulation

4.1 Introduction

Previous measurements of sound speed and attenuation on whole blood coagulation have shown that coagulation process may be affected by temperature, hematocrit, and thrombin or calcium chloride (Shung et al. 1984; Ossant et al. 2004; Huang et al. 2005). This observation can be interpreted that the sound speed and attenuation coefficient of human blood coagulation at real body temperature could be monitored in real time at a certain hematocrit and natural process without any chemicals.

The investigation of acoustic properties from whole blood during coagulation as a function of time at body temperature of 37°C is of interest because coagulation process depends on temperature. Since natural blood coagulation occurs without any chemicals, no chemicals are added. For these reasons, the physiological changes of blood coagulation in terms of sound speed and attenuation were investigated from whole blood after exposed to the air without adding any chemicals. The sound speed and attenuation coefficient of whole blood coagulation was measured at a frequency of 5 MHz unfocused transducer at 37°C.

4.2 Blood preparation

Experiments were conducted using 7 ml of whole blood of ten individual healthy volunteers, without any history of bleeding or thrombosis. The blood was drawn from a forearm vein of a healthy volunteer using a syringe needle. Blood from a syringe was immediately transferred to the sample container for measuring the acoustic properties of whole blood. For each sample, all experiment set up was arranged before drawing the blood.

4.3 Method for whole blood coagulation measurement

Experimental set up was shown in Fig 3.2. The test sample container was placed perpendicular to the direction of propagation of ultrasound. The temperature of water bath was adjusted to 37°C. The distance between the face of the transducer and the sample was adjusted. The correct level was initially determined using amplitude of the first echo from the front face of sample container and perfect reflector was maximized on the oscilloscope. Time was allowed for the temperature to be stabilized. The blood was drawn on site by an experienced nurse through a 10 ml syringe from a healthy volunteer at the test place. It was immediately transferred to the sample container within few seconds. Broadband pulses were transmitted at a pulse repetition frequency of 500 Hz. The reflected signals from the blood sample and a steel reflector were digitized at a sampling rate of 250 MHz. Data were collected in every 4 seconds over 65 minutes. The data were analyzed using Matlab software. The whole process was shown in block diagram as shown in Fig.4.1.



Figure 4.1 Block diagram of whole blood coagulation process

4.4 Results

The temporal variations of sound speed and attenuation coefficient during blood coagulation at 37°C are presented in Figs. 4.2,4.4,4.5 & 4.7. The ten curves (A, B, C, D, E, F, G, H, I, J) with different markers in the figures represent sound speed and attenuation coefficient variations of ten individuals blood samples during coagulation. Out of ten blood samples, eight samples showed the same trend of curves and the other two curves show different pattern. These two samples are not consistent with the other results. These variations may be not from the measurement errors but from the differences of blood itself, since the attenuation coefficients are similar to the ones from the other blood samples. It requires further investigation. Sound speed and attenuation variations from two samples were shown separately in Figs. 4.4 & 4.7 respectively. In this research, only *eight* blood samples were considered for further explanation. The average sound speed of eight blood samples over an hour during coagulation was increased from 1585 m/s to 1596 m/s as shown in Fig 4.3. Lines and error bars with open circles in the Fig. 4.3 represent the average and the standard deviation of eight sample measurements, respectively. These results were in good agreement with the previous ones (Shung et al. 1984; Voleisis et al. 2002). Table 1 showed the average sound speed and attenuation coefficient of whole blood during coagulation and the results of the other groups were given for comparison.

The curves represent the peculiar stages of blood coagulation of volunteers in Fig. 4.2. At the beginning stage of coagulation process up to 5

minutes, sound speed of whole blood sharply increased. The slope became less steep after 5 minutes, and further changed at around 27 minutes. After that time, sound speed from *seven* blood samples was not changed much. The other one showed the different pattern from the other *seven* samples, though the sound speed variations are similar.

Attenuation coefficient of eight individual blood samples during coagulation was shown in Fig. 4.5 and their patterns are different from the ones of sound speed. Average attenuation coefficient of eight blood samples was increased from 2.5 dB/cm to 4.3 dB/cm at 5 MHz as shown in Fig 4.4. Lines and error bars with open circles in the figures represent the average and the standard deviation of eight sample measurements, respectively. At the beginning stage of coagulation process, the attenuation of eight blood samples was decreased or remained the same up to about 12 minutes, while the sound speed was increased. At that time it showed minimum attenuation of 2 dB/cm (Fig 4.6). After 12 min, it increased almost linearly with time over 65 minutes.

제주대학교 중앙도서관 JEJU NATIONAL UNIVERSITY LIBRARY

29

| Parameters | | Previous researchers | |
|------------------------------------|-------------------------------------|----------------------------------|----------------------------------|
| | Whole blood at 37 ⁰ C | K.K.Shung et al. | A.Volesis et al. |
| Acoustic properties | | Whole blood at 23 ⁰ C | Whole blood at 37 ⁰ C |
| Sound speed (m/s) | 1585-1596 제주대 | ¹⁵⁵⁶⁻¹⁵⁹⁰ 학교 중앙도서관 | 1585-1600 |
| Attenuation coefficient (dB/cm) | 2.5-4.3 | 1.7 - 4.0 | |

Table 1 Average sound speed and attenuation from eight samples
during blood and plasma coagulation and comparison with the
previous groups



Figure 4.2 Sound speed variations of eight individuals during blood coagulation



Figure 4.3 Average sound speed variation of eight individuals during blood coagulation



Figure 4.4 Sound speed variation of nine and ten individual during blood coagulation



Figure 4.5 Attenuation coefficient variation of eight individuals during blood coagulation



Figure 4.6 Average attenuation coefficient variation of eight individuals during blood coagulation



Figure 4.7 Attenuation variation of nine and ten individual during blood coagulation

4.5 Discussion

Whole blood coagulation processes were investigated in terms of acoustic properties. Sound speed and attenuation coefficient were increased during blood and plasma coagulation processes, but they changed with the different patterns. Sound speed variation changed in three phases. At first, sound speed increased sharply until about 5 minutes, followed by a gradual increase until 27 minutes and then slow change toward a steady state situation. The variation of sound speed seemed to correspond to the three physiological phases of coagulation, the enzymatic phase, the propagation phase, and the termination phase. However attenuation variation did not show the three phases. Attenuation was stable or decreased until 12 minutes and then increased linearly with time.

In the first phase, sound speed of eight blood samples started at different values but increased rapidly up to 2 m/s within 5 minutes (Fig. 4.2). At this phase, soft and gel type coagulum was formed as prothrombin converted into thrombin in both intrinsic and extrinsic coagulation pathways. After that the sound speed increased progressively, while attenuation was stable or decreased until 12 minutes. The changes of attenuation coefficient are from the changes in acoustic impedance of the clotted blood, mainly because of an increase in the reflection and the scattered energy losses (Shung et al. 1984). At this phase the coagulated blood was in elastic medium and the energy loss in elastic medium is less. The slope of sound speed and attenuation is dependent on the strength of the coagulum, which may signifies the beginning of proper coagulation because of the bulk of thrombin formation and the reaction with fibrinogen (Mann 2002). In the second phase, the sound speed showed a gradual increase until 27 minutes and sound speed of eight samples was nearly equal to the same value

(~1593 m/s) at about 27 minute which was indicated in small error bar in Fig.4.3. Attenuation was increased linearly from 2~4.5 dB/cm during this stage because the viscosity of blood increased (Wang et al. 2000; Wang and Tsui 2004). The increase of attenuation and sound speed of eight blood samples may indicate the possibility of solid fibrin polymer formation from fibrinogen (Shung et al. 1984).

The third phase of whole blood coagulation seemed to start after formation of solid fibrin, i.e. after 27 minute of coagulation. Average sound speed was increased 3 m/s smoothly until 65 minutes in whole blood. Seven curves showed almost the similar patterns of variations except one showing higher sound speed, while attenuation kept increasing until 65 minutes. This phase may be because of activation of protein C with cofactor protein S and propagation of reactions with thrombomodulin within the blood sample (Mann 1992; Comp 1982).

4.6 Conclusion

The effects of blood coagulation on the acoustic properties from ten volunteers were investigated. It was found that sound speed and attenuation of blood increased during coagulation. The variation of sound speed during whole blood coagulation showed the three distinguishable changes of slope with respect to time, and these three different variations may be comparable to the physiological phases (enzymatic, propagation and termination phases) of blood and plasma coagulation. The processes of whole blood coagulation at nearly body temperature were complicated and sound speed and attenuation fluctuations are relatively high.

Chapter V

Measurement of acoustic properties of plasma during coagulation

5.1 Introduction

In this chapter, the whole process of plasma coagulation was measured in terms of sound speed and attenuation coefficient as a function of time. Ultrasonic backscatter signals from spherical glass particles (Alves and Machado 1991) and ultrasonic shear wave method (Alves and Machado 1994) have been employed to obtain the plasma clotting time. Ultrasonic measurements of the whole process of plasma coagulation with aPTT were not yet reported. The structural changes of whole process of plasma coagulation with aPTT as function of time were measured as temporal changes of sound speed and attenuation this research.

5.2 Plasma preparation 제주대학교 중앙도서관

Experiment was performed using 2 ml of plasma of ten individual healthy volunteers, without any history of bleeding or thrombosis. 6 ml of blood was drawn from a forearm vein of a healthy volunteer using a syringe needle. After discarding the first milliliter, 90% volume of blood was collected in a polystyrene tubes containing 10% volume of 3.8% trisodium citrate. The blood was centrifuged for 15 min at about 3500 RCF (relative centrifugal force) using a centrifuge (Hanshin Medical Co. Ltd., Model-AT-650) and then plasma was separated and stored in -70°C until assayed.

5.3 Method for plasma coagulation measurement

Experiment was performed with plasma samples of ten individuals. The experimental set up was shown in Fig. 3.2 and the same as used in the measurement of the blood coagulation, as explained in chapter 4. For all plasma sample measurements, 2 ml of plasma was placed into a sample container using a pipette and incubated at 37±0.2°C for not longer than 5 minutes to avoid loss of factors V and VII. Then 2 ml of aPTT was added and thoroughly mixed by using a magnetic stirrer. After incubation for 5 minutes to reach thermal equilibrium, 2 ml of pre-incubated calcium chloride (0.02M) solutions was added for initiation of coagulation. Ultrasonic broadband pulses were transmitted at a pulse repetition frequency of 500 Hz. The reflected signals from the plasma sample and a steel reflector were digitized at a sampling rate of 250 MHz. Data were collected in every 4 seconds over 45 minutes. Post-processing was done in Matlab software. The block diagram of whole process of plasma coagulation was as shown in Fig. 5.1.





Figure 5.1 Block diagram of plasma coagulation process

5.4 Results

Sound speed and attenuation variations of ten individuals during plasma coagulation at 37^oC are presented in Figs 5.2, 5.3, 5.5 & 5.7. The curves (A, B, C, D, E, F, G, H, I, J) with different markers in the figures represent the structural changes of ten plasma samples during coagulation. In case of plasma, eight samples out of ten plasma samples, showed the same trend of curves and the other two curves showed the different pattern. These two samples variations are not consistent with the other result. These variations may be not from the measurement errors but from the differences of plasma itself, since one sample sound speed and the other sample attenuation coefficient are similar to the ones from the other plasma samples. It requires further investigation. Sound speed and attenuation variations of two samples were shown separately in figures 5.4 & 5.7 respectively. In this research, only *eight* plasma samples were considered for further explanation. The average sound speed of *eight* plasma samples over 45 min of coagulation was increased from 1535 m/s to 1542 m/s which was shown in Fig 5.3. Lines and error bars in the figures represent the average and the standard deviation of eight samples, respectively. The plasma coagulation process is significantly faster compared to the blood coagulation. At initial phase the sound speed was increased rapidly over about 5 minutes. Between 5 to 27 minutes, it increased slowly and after that it remained the same. Sound variations among different plasma samples are smaller excluding one sample (A) of higher sound speed.

The variation of ten individuals attenuation coefficient of plasma during coagulation is shown in Figs. 5.5 & 5.7. Figure 5.6 shows the average and the standard deviation values of the relative attenuation coefficient of plasma during coagulation as a function of time. Average attenuation

coefficient of eight plasma samples was increased form 1.2 dB/cm to 3.1 dB/cm. Attenuation coefficient was linearly increased over 45 minutes except at an early stage of a few minutes. The variation among the different samples was small, it was indicating clearly by small error bars (Fig. 5.6). It was increased linearly even after the sound speed was not changed.





Figure 5.2 Sound speed variation of eight individuals during plasma coagulation



Figure 5.3 Average sound speed variation of eight individuals during plasma coagulation



Figure 5.4 Sound speed variation of nine and ten individual during plasma coagulation



Figure 5.5 Attenuation coefficient variation of eight individuals during plasma coagulation



Figure 5.6 Average attenuation coefficient variation of eight individuals during plasma coagulation



Figure 5.7 Attenuation variation of nine and ten individual during plasma coagulation

5.5 Discussion

Sound speed of plasma also showed three phases of coagulation. Plasma coagulation process was fast compared to whole blood and not much varied among the samples. In the first phase, sound speed and attenuation of eight individuals are different from each other and sound speed was increased rapidly up to around 4 m/s within 5 minutes in Fig. 5.2. However, attenuation was decreased or stable, probably because sound speed depends on bulk modulus and density. For plasma coagulation, initial enzymatic reactions during intrinsic and extrinsic pathways made the coagulum soft and stable. In second phase, the sound speed increased slowly for 7 subjects (except curve "A" in Fig. 5.2). Sound speed reached the same value (~1540m/s) at about 27 minute but attenuation increase of sound speed and linear increase of attenuation may indicate the possibility of thrombin reaction with fibrinogen and formation of fibrin.

The third phase of plasma coagulation seemed to start after formation of solid fibrin, i.e. after 27 minute of coagulation; sound speed was increased smoothly in 45 minutes in plasma. Seven curves showed almost the similar patterns of variations except one showing higher sound speed, while attenuation kept increasing linearly even after sound speed reaches to the same value. This phase may be because of activation of protein C with cofactor protein S and propagation of reactions with thrombomodulin within the blood sample (Mann 1992; Comp 1982).

Comparing Fig.4.2 with Fig.5.2, the responses of sound speed of blood and plasma were different during formation of clot. Blood coagulation is more complex and slower than plasma coagulation. From Figs. 5.2 and 5.5, sound speed of plasma changed faster only at the initial stage of the clotting

process, while attenuation was increased linearly with time even after sound speed was not changed. Simultaneous monitoring of sound speed and attenuation may give a better understanding of the process of coagulation. Sound speed and attenuation curves of J & I shown in Figs. 5.4 & 5.7, respectively revealed that variations are different from the other curves because the rate of conversion of fibrinogen to the insoluble product fibrin or activation of blood coagulation factors may be different from the others. Attenuation coefficient of plasma was increased almost linearly as shown in Fig. 5.6. For plasma coagulation, an amount of aPTT and calcium chloride at the body temperature have impacts on determination of the rate of coagulation. Coagulation process is also dependant on the temperature and the coagulation factors of blood, so all experiments have been performed in the body temperature. For plasma coagulation, clinical laboratories usually calibrate the procedures and equipments for coagulation test, and aPTT is usually in the range of 24-32 sec. Deviation from the aPTT time indicates that some coagulation factors malfunctioned in the process. It needs to be resolved in the future and may give more information about blood clotting process. More works are needed in order to understand the mechanisms of the change of sound speed and attenuation during the coagulation of plasma. Measurements in a wide frequency range and as a function of temperature with more samples may give additional information on coagulation process of blood and plasma.

5.6 Conclusion

The effects of plasma coagulation on the acoustic properties from ten volunteers were investigated. It was found that sound speed and attenuation of plasma increased during coagulation. Unlike the aPTT time alone, the monitoring of sound speed and attenuation during whole process of plasma coagulation using ultrasound may give additional information. The variation of sound speed during plasma coagulation also showed the three distinguishable changes of slope with respect to time, and these three different variations may be comparable to the physiological phases of plasma coagulation. Attenuation of plasma increased linearly even after the sound speed was not changed. It may be because of some factors are malfunctioned with in the sample.



Chapter VI

Measurement of acoustic properties of plasma during coagulation with Heparin

6.1 Introduction

Plasma coagulation with an anti-coagulant was studied in this chapter. Commercially available brand of heparin was selected to study the effects of specific factors and process of coagulation. It has been found that heparin is an effective antithrombotic agent that reduces morbidity and mortality among the patients with deep vein thrombosis, pulmonary embolism and also prevents treatment of cardiovascular disorders. Patients are commonly administered by heparin to minimize the risks associated with surgical procedures. The activated partial thromboplastin time (aPTT) and the activated clotting time (ACT) are the most extensively applied measurements of the heparin effect. Accurate evaluation of the effect of heparin is crucial because of the heparin responses vary greatly among the individuals. A monitoring system of whole process of coagulation must be established.

A coagulation analyzer was frequently used to quantify the perfusion of heparin as a function of aPTT or ACT time format in clinical tests (Cheng et al.1998, Hepel et al.1999). It provides results within minutes so that rapid adjustment of heparin therapy can be made. However, these instruments are unable to monitor the whole process of heparin therapy in diagnostics. Therefore, ultrasonic method was used to measure sound speed and attenuation coefficient variations in plasma during coagulation with heparin.

The physical properties of plasma coagulation with heparin have been studied in terms of sound speed and attenuation as a function of time, the effects of plasma coagulation without and with heparin have also been investigated. The measurements verified that the sound speed and attenuation variations are significantly affected by heparin during coagulation process.

6.2 Method for plasma coagulation with heparin measurement

Experiment was performed with plasma samples of five individuals. The experimental set up was shown in Fig. 3.2 and the same as used in the measurement of the blood coagulation, as explained in chapter 4. For all plasma coagulation with heparin measurements, 0.9 ml of plasma was placed in to the sample container using 1 ml capacity pipette and incubated at 37±0.2°C for not longer than 5 minutes to avoid loss of factors V and VII. Then 0.1 ml of 100 µg/ml concentrated heparin was added using 20~200 µl capacity pipette. After 1 minute 1 ml of aPTT was added and this mixture was thoroughly mixed by using a magnetic stirrer and incubated for 5 minutes to reach thermal equilibrium. After that 1 ml of pre-incubated calcium chloride (0.02M) solutions was added for initiation of coagulation. Ultrasonic broadband pulses are transmitted at a pulse repetition frequency of 500 Hz. The reflected signals from plasma sample and steel reflector were digitized at a sampling rate of 100 MHz using NI high speed digital oscilloscope (PCI-5122, National instrument, Austin, TX, USA). Data were collected every 4 seconds over 30 min. collected data were analyzed using LABVIEW software to obtain plasma profiles in real time. The block diagram of whole process was as shown in Fig 6.1.



Figure 6.1 Block diagram of plasma coagulation with heparin process

6.3 Results

Sound speed and attenuation coefficient variations of five individuals during plasma coagulation at 37°C are presented in Figs 6.2 & 6.4. The curves (A, B, C, D, and E) with different markers in the figures represent the structural changes of five plasma samples with heparin during coagulation. The average sound speed of five plasma samples was increased 1 m/s at initial stage of 5 minutes after that it reached to steady state. The peak value of sound speed was showed after 3 minutes of coagulation (Fig. 6.3), after that it slightly decreased and reached to steady state at about 5 min. From 5 minutes to over 10 minutes the sound speed was not changed as shown in Fig. 6.2. Lines and error bars with open circles in Fig. 6.3 represent the average and the standard deviation of five measurements, respectively. Sound speed variations among different plasma samples are smaller.

Attenuation coefficient of plasma coagulation with heparin slightly increased for first few seconds after that it was observed to be stable (fig 6.4). Fig 6.5 shows the mean and the standard deviation value of relative attenuation coefficient parameters of five plasma samples measurements, respectively. Attenuation coefficient of plasma coagulation with heparin did not show any significant variation.



Figure 6.2 Sound speed variation of five individuals during plasma coagulation with heparin



Figure 6.3 Average sound speed variation of eight individuals during plasma coagulation with heparin



Figure 6.4 Attenuation coefficient variation of five individuals during plasma coagulation with heparin



Figure 6.5 Average attenuation coefficient variation of five individuals during plasma coagulation with heparin
6.4 Discussion

The present study measured the acoustic properties of plasma coagulation with heparin and then investigated anticoagulant activity of heparin *in vitro* with aPTT as a function of time. Figs. 6.2 & 5.2 represent the behaviour of sound speed variations of plasma coagulation with and without heparin. Sound speed variations of plasma with and without heparin assays during coagulation were different from each other, but their patterns were similar. In case of plasma without heparin contained assay showed higher sound speed nearly 6 m/s and also attenuation increased linearly from 1.25 dB/cm to 3.1 dB/cm over 45 minutes. In case of plasma with induced heparin showed at initial stage of sound speed increased only 1 m/s until 5 minutes (Fig. 4) after that sound speed fall down to a relatively low value and then attained steady state. Attenuation also not changed much except beginning of few seconds.

These methods were compared in determining the heparin anticoagulant activity. In case of plasma coagulation without heparin, sound speed kept increased over the observation time. Where as, the sound speed would gradually dropped to relatively low value at about 300 seconds in case of heparinized plasma. Apparently, in vitro, sound speed of plasma with heparin disappears faster than that of without heparin. In general, it would take 15 minutes for normal plasma coagulation, owing to a series of cascade enzymatic reactions, to take place.

In fact, the time of decrease in sound speed responses was extremely dependent on heparin concentration. Obviously, heparin anticoagulant activity works at initial phase of coagulation according to physiological process. It could be concluded that the effects of heparin can work only at initial enzymatic phase of plasma coagulation. Simultaneous monitoring of

sound speed and attenuation of plasma coagulation with and without anticoagulant may give a better understanding of anticoagulant activity of anticoagulant. Comparing Fig.6.2 with Fig.5.2, the responses of sound speed of plasma with and without heparin were different during formation of clot. Plasma coagulation with heparin is harder and slower than plasma coagulation without any heparin. From Figs. 5.6 and 6.5, plasma without heparin assay indicating that attenuation coefficient was increased linearly with time even after sound speed remained the same for plasma without heparin. However, attenuation coefficient was not changed throughout the process in heparinized plasma. It could be concluded that heparin may affect the malfunctioning factors with in the sample.

6.5 Conclusion

The application of ultrasonic pulse echo method to study the plasma coagulation with an anticoagulant, such as heparin has been demonstrated. It was the first time has been used to measure the plasma coagulation with heparin using ultrasound. Sound speed variation had an apparent step ladder curves that was distinctly suitable for determination the coagulation time. Sound speed and attenuation variations of plasma without and with anticoagulant are different from each other. Plasma coagulation with heparin showed only at initial phase of reactions, sound speed and attenuation fluctuations were lower compared to plasma coagulation without heparin. Monitoring of plasma coagulation without and with heparin may give more information of anticoagulant activity of an anticoagulant. These results might help a better understanding of coagulation process, which may be applied in various techniques for development of anticoagulant and evaluation of the drugs.

Chapter VII

Conclusions and Future work

7.1 Conclusions

The effects of blood and plasma coagulation on the acoustic properties from ten volunteers were investigated. Further plasma coagulation with heparin as an anticoagulant was also investigated. It was found that sound speed and attenuation of blood and plasma increased during coagulation. Unlike the aPTT time alone, the monitoring of sound speed and attenuation during whole process of blood and plasma coagulation using ultrasound may give additional information. The variation of sound speed during whole blood and plasma coagulation showed the three distinguishable changes of slope with respect to time, and these three different variations may be comparable to the physiological phases (Enzymatic, Propagation and Termination phases) of blood and plasma coagulation. The processes of whole blood coagulation at nearly body temperature with natural process were complicated and sound speed and attenuation fluctuations are high compared to plasma coagulation. Sound speed and attenuation variations of whole blood and plasma are different from each other. Sound speed and attenuation of whole blood were increased over 60 minutes, but sound speed of plasma rapidly increased in the first stage and then remained similar while attenuation increased linearly over 45 minutes.

Sound speed and attenuation variations of plasma without and with an anticoagulant are different. Plasma coagulation with heparin changed only at initial phase of reactions and sound speed and attenuation fluctuations are

smaller compared to plasma coagulation without heparin. Finally plasma coagulation with an anti-coagulant such as heparin was found helpful to study the effects of the anti-coagulant on specific factors and/or processes of coagulation. These results might help a better understanding of coagulation processes, which may be applied in various techniques for development of anticoagulants and drugs.

7.2 Future work

Extending this work to measure the backscattering from whole blood during coagulation is required, since it may give more information to understand the phases of coagulation. In addition to precise measurements of the backscattering from blood during coagulation, B-mode or harmonic imaging may be directly applied to measure the blood coagulation inside of blood vessel in real time. Combining the sound speed and attenuation with scattering measurements would yield to implement the diagnosis and therapy in cardiac surgery process, deep vein thrombosis and pulmonary embolisms.

Another approach is to measure the anti-coagulant activity of heparin or other anticoagulants may help to implement anticoagulants and development of the drugs. In appendix, the preliminary experiments with other anticoagulants from seaweed were presented.

BIBLIOGRAPHY

Alives CHF, Machado JC. Measurement of plasma coagulation time using ultrasound shear waves. Physiol Meas 1994; 15:309-316.

Björn Dahlbäck. Blood coagulation. The Lancet 2000; 355:1629-1632.

- Brown BA. Hematology: Principles and Procedures, 4th ed.Lea and Febiger.Philadelphia 1984.
- Brummel KE, Paradis SG, Butenas S, Mann KG. Thrombin function during tissue factor induced blood coagulation. Blood 2002; 100:148-152.
- Brummel KE, Butenas S, Mann KG. An integrated study of fibrinogen during coagulation. The Journal of Biological Chemistry 1999; 274: 22862–22870
- Bussey HI. Problems with monitoring heparin anticoagulation. Pharmacotherapy 1999; 19(1):2-5.
- Butenas S, Mann KG. Blood coagulation. Biochemistry (Moscow), 2002:67(1): 3-12. 지주대학교 중앙도서관
- Butenas S, Vant veer C, Mann K. Normal thrombin generation. Blood 1999; 94:2169-2178.
- Carstensen E, Kam LI, Schwan HP. Determination of the acoustic properties of blood and its components. J.Acoustic.Soc Am 1953; 25:286-289.
- Carlson JE, Deventer JV, Micella M. Accurate temperature estimations in ultrasonic pulse echo systems. WCU 2003; 7-10:1563-1568.
- Chang HC, Cheng TJ, Wu TH, Lin TM. Determination of coagulation time in whole blood containing anticoagulation by piezoelectric quartz crystal sensor. Sensors and actuators 2000; B66:296-298.

- Cheng TJ, Chang HC, Lin TM. A piezoelectric quartz crystal sensor for the determination of coagulation time in plasma and whole blood, Biosens. Bioelectron. 1998; 13: 147–156.
- Comp PC, Acocks RM, Ferrell GL, Esmon CT. Activation of protein C in vivo. J.Clin.Invest 1982; 70:127-134.
- Davey FR, Fiske ML, Malley O. Evaluation of a photoelectric automatic prothrombin analyzer. Am.J.Clin. Path. 1972; 58:687-692.
- Dacie and Lewis Practical Hematology, Edited by S Lewis Barbara, 9th Edition
- Daniel JM, Toce JA, Anevski PJ, Tollefsen DM et al. Anticoagulant and antithrombotic activity of maitodapoh, a novel sulfated tetrasaccharide. The Journal of Pharmacology and Experimental therapeutics 1999; 288:516-521.
- Esmon CT. The role of protein C and thrombomodulin in the regulation of blood coagulation. The Journal of Biological chemistry 1989; 264: 4743-4746.
- Gentry PA. Comparative aspects of blood coagulation. The veterinary Journal 2004; 168,238-251.
- Grosso VAD, Mader CW. Speed of sound in pure water. J.Acoustic.Soc Am 1972; 52:1442-1446.
- Grybauskas P, Kundrotas K, Sukackas V, Yaronis E. Ultrasonic digital interferometer for investigation of blood clotting. Ultrasonics 1978; 16:33–36.
- Guyton AC, Hall JE. Hemostasis and blood coagulation. Textbook of Medical Physiology. Tenth Edition. W.B. Saunders Co. Philadelphia 2000.

- Hepel M, Tewksbury E. Ion-gating phenomena of self-assembling glutathione films on gold piezoelectrodes, J. Electroanal. Chem 2003; 552: 291–305.
- Hepel H, Tewksbury E. Nanogravimetric study of templated copper deposition in ion-channels of self-assembled glutathione films on gold piezoelectrodes, Electrochim. Acta 2004; 49: 3827–3840.
- Hepel H. Electrode-solution interface studied with electrochemical quartz crystal nanobalance, in: A. Wieckowski (Ed.), Interfacial Electrochemistry – Theory, Experiment and Applications, Marcel Dekker, New York 1999, 599–630.
- Huang CC, Wang SH, Tsui PH. Detection of blood coagulation and clot formation using quantitative ultrasonic parameters. Ultrasound in Med. & Biol 2005; 31:1567-1573.
- Jacobs JE, Malinka AV, Haque P, Jhabvala MD. Ultrasound spectroscopy applied to blood coagulation studies. Ultrasonics 1976; 3:84–90.
- Kažys R, Voleišis A, Voleišiene B, Mažeika L et al. Ultrasonic multichannel method for investigation of non-stationary biologic liquids. Ultragarsas 2003; 4(49):53-57.
- Kinsler L B, Frey A R, Coppens A B and Sanders J V. Fundamentals of Acoustics, 4th ed; John Wiley &Sons, New York.2000;152-155.
- Kuo IY, B.Hete B, Shung KK. A novel method for the measurement of acoustic speed. J.Acoustic.Soc Am 1990; 88 (4):1679-1682.
- Machado JC, Lenzi A, Silva WG. An ultrasonic method to measure human plasma coagulation time J.Acoust.Soc.Am 1991; 90 (4):1749-1753.
- Madsen LEL, Dong F, Frank GR, Garra BS. Interlaboratory comparison of Ultrasonic backscatter, attenuation and speed measurements, American institute of ultrasound in medicine. J Ultrasound Med 1999;

18:615-631.

- Mann KG.Normal hemostasis.In:WN.Kelley, ed. A text book of internal Medicine. Philadelphia, PA: Lippincott; 1992:1240-1245.
- Mann KG, Butenas S, Brummel K. The dynamics of thrombin formation. Arterioscler Thromb Vasc Biol.2003; 23:17-25.
- Maria P, McGee S, Lynne C. Functional Difference between Intrinsic and Extrinsic Coagulation Pathways .The Journal of Biological Chemistry 1991;266:6079-6085.
- Miale B.The fibrometer system for routine coagulation tests. Am.J.Path 1965; 43:475- 480.
- Milstone JH. Three stage analysis of blood coagulation. The Journal of General Physiology 1947; 301-324
- Mongirdiene A. Comparison data of blood coagulation process, investigated by ultrasonic impulsive coagulograph and electrocoagulometer. Ultragarsas 1999; 2(32):36-37.
- Ossant F, Libgot R, Coup P, Lermusiaux P, Patat F. High frequency ultrasound characterization of the coagulation process of whole blood. IEEE Ultrasonic Symp Proc 2004; 2:846–849.
- Pieiere Y, Cariou J, Y.Guern Y, Cariou GL et al. Evaluation of blood plasma coagulation dynamics by speckle analysis. Journal of biomedical Optics 2004; 9 (2):408-412.
- Shanmugam M, Mody KH. Heparinoid activity sulphated polysaccharides from marine algae as potential blood anticoagulant agents. Current science 2000; 79:1672-1683.
- Shung KK, Fei DY, Yaun YW, Reeves WC et al.Ultrasonic characterization of blood during coagulation. J.Clin Ultrasound 1984; 12:147-153.

- Shung KK, Sigelmann RA, Schmer G. Ultrasonic measurement of blood coagulation time. IEEE Trans Biomed Eng. 1975 Jul; 22(4):334-7.
- Sobas FH, French F, Trzeciak MC, Negrier C et al. Human plasma fibrinogen measurement derived from activated partial thromboplastin time clot formation. Blood coagul Fibrinolysis 2002; 13(1): 61-68.
- Stride E, Saffari N. Theoretical and experimental investigation of the behaviour of ultrasound contrast agent particles in whole blood. Ultrasound in Med. & Biol., 2004; 30:1495-1509.
- Voleišis A, Kažys RJ, Mažeika L, Šliteris R et al. Ultrasonic technique for the investigation of structural properties of biological fluids. J Acoust Soc Am 1999; 105:1369 –1370.
- Voleišis A, Kažys R, Mažeika L, et al. Ultrasonic method for the whole blood coagulation analysis. Ultrasonic 2002; 40:101–107.
- Voleišiene B, Voleišis A, Kažys RJ, Mažeika L, Šliteris R. Dynamic calibration method for ultrasonic coagulometry. Ultragarsas 2002; 3(44):45-49.
- Wang YH, Chung TW, Lai JY, Hsu JJ, Lin YS. A viscometric method to study the effects of hematocrit of blood and different surfaces of biomaterials on blood clot formation. J Chin Inst Chem Engrs 2000; 31:27–32.
- Wang SH, Lee LP, Lee JS, A linear relation between the compressibility and density of blood. J.Acoust.Soc.Am 2001; 109(1):390-396.
- Wang SH, Tsui PH. Ultrasonic techniques to assess the properties of hard and soft biological tissues. Key Eng Mater 2004; 270273:2055–2060.
- Yashantha A.Anticoagulative and antihypertensive activity of ecklonia cava. Master thesis, Cheju National University, Cheju, 2004.

APPENDIX

Measurement of acoustic properties of plasma during coagulation with brown and red algae

A.1 Introduction

Heparin and Warfarin are the sole anticoagulants used to prevent thrombotic disorders and treatment of cardiovascular disorders. Heparin and its derivatives have low molecular weight, although an effective antithrombotic agent that reduces morbidity and morality among patients with deep vine thrombosis and pulmonary embolism. However, it cannot be administered orally and is therefore difficult to maintain the extended therapy (Daniel et al. 1999). Warfarin is an orally bioavailable anticoagulant that inhibits production of vitamin k dependent coagulation factors. However the anticoagulant agent delays onset and offset of action. Therefore it takes several days to reach therapeutic anticoagulant levels, requires frequent monitoring of anticoagulation and is subject to many drug interactions (Daniel et al. 1999). Due to these limitations, it is needed to develop new anticoagulants that exhibit rapid and predictable onset and offset kinetics.

Anticoagulant properties of marine algae have been extensively studied for last 60 years. Anticoagulants activity of marine algal polysaccharide has potential ability to prolong blood coagulation time. This is due to their hemiester helphate groups in their sugar residues (Shanmugam and Mody 2000). Therefore, seaweeds are good alternative sources for anti-coagulant drug production. In last decades, seaweeds or their extracts from red alga and brown alga have been studied (Shanmugam and Mody 2000) as novel

sources which have been reported to possess bioactivity of potential medicinal value.

In this study, an ultrasonic method was introduced for monitoring the change of whole process of plasma coagulation with anticoagulant from brown and red algae, but did not investigate whole process and anticoagulant activity.

A.2 Materials and Method

Two kinds of anticoagulants such as red alge and brown alga were obtained from marine brown algae, *Ecklonia cava*. This *E.cava* was collected from sea shores of Jeju Island in Korea and processed (Yashantha 2004) in Marine biology laboratory at Cheju National University.

Experiment was performed with plasma samples of five individuals. The experimental set up was shown in Fig. 3.2 and the same as used in the measurement of the plasma coagulation with heparin, as explained in chapter 6. For all plasma coagulation with red alge or brown alge measurements, 0.9 ml of plasma was placed in to sample container using 1 ml capacity pipette and incubated at 37 ± 0.2 °C for not longer than 5 minutes to avoid loss of factors V and VII, and then 0.1 ml of 100 µg/ml concentrated red algae or brown algae was added using 20~200 µl capacity pipette. After 1 minute 1 ml of aPTT was added and this mixture was thoroughly mixed by using a magnetic stirrer and incubated for 5 minutes to reach thermal equilibrium . After that 1 ml of pre-incubated calcium chloride (0.02M) solutions was added for initiation of coagulation.

A.3 Results and Discussion

Sound speed and attenuation variations of five individuals during plasma coagulation with brown and red algae compounds at 37°C are presented in Figs 8.1, 8.3, 8.5 & 8.7, respectively. The curves (A, B, C, D, E,) with different markers in the figures represent the variations of five individuals plasma samples during coagulation. Lines and error bars in the Figs.8.2, 8.4, 8.6 & 8.8 represent the average and the standard deviation of sound speed and attenuation of five individuals plasma coagulation with brown algae and red algae, respectively.

Sound speed variation of five individuals during plasma coagulation with brown algae was increased nearly 1 m/s until 7 min after that it seams to be stable. There is not much variation among the samples as seen in the small error bars in Fig. 8.2. Attenuation coefficient was observed to be stable during coagulation.

Sound speed of five plasma samples during with red algae did not show any significant variations. The sound speed seems to be increased during first 2 min (Fig.8.6) but variation of the samples is pretty big and the results were not consistent to identify differences among the samples. Sound speed during plasma coagulation with red algae was fluctuated too much compared to plasma coagulation with brown algae. However, attenuation coefficient was similar to the one with brown algae (Fig.8.7).



Figure 8.1 Sound speed variation of five individuals in plasma with brown algae during coagulation



Figure 8.2 Average sound speed variation of five individuals during plasma coagulation with brown algae



Figure 8.3 Attenuation variation of five individuals during plasma coagulation with brown algae



Figure 8.4 Average attenuation variation during plasma coagulation with brown algae



Figure 8.5 Sound speed variation of five individuals during plasma coagulation with red algae



Figure 8.6 Average sound speed variation of five individuals during plasma coagulation with red algae



Figure 8.7 Attenuation variation of five individuals during plasma coagulation with red algae



Figure 8.8 Average attenuation of five individuals during plasma coagulation with red algae

According to these results it is difficult to understand the brown algae and red algae activity to any specific factors or process. These results were similar to results of plasma coagulation with heparin, as explained in chapter 6.

A.4 Conclusion

The application of ultrasonic pulse echo method to study the plasma coagulation with anticoagulants from brown and red algae has been demonstrated. It was the first time has been used to measure variations of plasma coagulation with brown and red algae using ultrasound. The responses of sound speed variation may not be consistent to find specific process of coagulation or prolonged clotting time. One finding was that the attenuation coefficient was stable in both cases during coagulation. Sound speed fluctuations were big among the samples compared to plasma coagulation with heparin. These results were may not be enough to explain the anti coagulant activity of brown and red algae and it is required to study further.

ACKNOWLEDGEMENTS

Journey is easier when you travel together. Interdependence is certainly more valuable than independence. This thesis is the result of TWO years of work in which many had their helping hand. It's a great opportunity to express my gratitude for all of them.

First of all, I would like to express my sincere appreciation and respect to my advisor Prof. Dong-Guk Paeng for his thoughtful guidance, friendly nature, nice encouragement and valuable guidance throughout the course of my Masters study accompanied by this research. I admire his research skills especially the way in handling the problems that helped me to coordinate the things efficiently.

I am very grateful to my committee members and department professors, Prof. Jinho Bae, Prof. Min Joo Choi, Prof. Il-Hyoung Cho, and Prof. Joon Young Kim. I thank them for their kindness, friendliness and all the help they have provided in this study. Their suggestions and support were very helpful and invaluable throughout this study.

I would like to express my sincere thanks to Prof. You-Jin Jeon and Mr. Yashantha Athukorala, Mr. Cui xiu ji, Miss. An Chun Shan, Mr. Kang Gwan Suk, Mr. Anil Kumar, Mr. Nam Kwon-Ho for their help in many ways to finish my work successfully.

My sincere thanks to Mr. Herath, Mr. Rohan, Mr. Neaz, Mr. Nam, Mr. Anil, Mr.Ying, Mr. Gwan Suk, Mr. Nalin, Mr. Yashantha, Mr. Seungwoo, Mr. Hyeok Jun for their help to provide blood samples and Miss. Oh Gum Hee for collecting blood samples. With out their cooperative outlook this work wouldn't have been possible.

Finally, Special thanks are extended to my fellow graduate students Mr. Anil Kumar, Mr.Ying Li, Miss.Min Jung Kim, Mr. Seungwoo Byun, and Mr. Hyukchun Kho who are very friendly and helped in conducting many experimental tests in this study. I impart my special appreciation to my Indian friends Mr. Sridhar Kilaru and Mr.Vijaya Ramu Dirisala who cherished and fancied my life here.

I wish to thank the Cheju National University Bio-Medical Engineering Research Laboratory, Marine Biotechnology Laboratory, the laboratory of clinical research and the BK21 for their technical and financial support.

I would like to express my deepest gratitude and love to my dear parents and sisters for their support. Without their love, understanding, support and encouragement, the completion of this study would have never become possible. I thank them for all their love and help they are giving me. I am very much thankful to almighty for pondering divine blessings on me

Last but not the least I am thankful to all my friends, relatives and teachers who stood by my side in tough times to combat difficulties and gave me moral strength which paved the way for me to pursue Masters in Korea.